Author's response to reviews

Title: A Cost-Consequences analysis of the effect of Pregabalin in the treatment of peripheral Neuropathic Pain under medical practice conditions in Primary Care settings.

Authors:

Ana Navarro (navarrosiguero@eresmas.com)
Maía T Saldaña (zatosaldana@yahoo.es)
Concepción Pérez (cperez.hlpr@salud.madrid.org)
Sandra Torrades (sandra.torrades@ebi2005.com)
Javier Rejas (marina.desalas@pfizer.com)

Version: 2 Date: 26 March 2010

Author's response to reviews: see over
March, 2010

Dear Editor,

I am enclosing the revised and modified version of the manuscript entitled: “A Cost-Consequences analysis of the effect of Pregabalin in the treatment of peripheral Neuropathic Pain under medical practice conditions in Primary Care settings” to be reviewed and considered for publication as an original article in the journal “BMC Neurology”.

The authors of this manuscript are: Ana Navarro, María T. Saldaña, Concepción Pérez, Sandra Torrades and Javier Rejas.

The authors of this manuscript state that:

- All of the authors have seen and agreed to the submitted version of the manuscript and with the listing of the authors.
- The material is original and unpublished previously.
- We also agree with the transfer of the copyright from the authors to the publisher.
- Author’s contribution: This was a collaborative work, and the authors worked closely each other. Ana Navarro, Mª Teresa Saldaña, Concepción Pérez, Sandra Torrades and Javier Rejas participated in the design of the study, interpretation of data and writing of the manuscript. Sandra Torrades participated in the analysis and interpretation of data and in the preparation of the manuscript. All authors were responsible for literature review and extraction of references.
- List of contents: nº of pages 24, nº of tables 5, nº of figures 2.
- This study was supported by an unrestricted grant of Pfizer Spain.

The corresponding author is as follows:
Dr. Ana Navarro
Centro de Salud de Atención Primaria “Puerta del Ángel”
Paseo de Extremadura 113-115
28022 Madrid, España
If you require any further information, please do not hesitate to contact me:

Sincerely,

Ana Navarro